Author’s response to reviews

Title: Health care workers' knowledge, attitudes and practices on tuberculosis infection control, Nepal

Authors:

Anita Shrestha (shres.aanie@hotmail.com)
Dipesh Bhattarai (dipeshbhattarai@hotmail.com)
Barsha Thapa (barsha35@gmail.com)
Prem Basel (prembasel11@gmail.com)
Rajendra Wagle (rajendrarwagle@gmail.com)

Version: 2 Date: 06 Oct 2017

Author’s response to reviews:

As suggested, we have made necessary English language correction and submitted a clean copy.